**Author's response to reviews**

**Title:** The effects of vitamin D2 or D3 supplementation on glycaemic control and related metabolic parameters in people at risk of type 2 diabetes: protocol of a randomised double-blind placebo-controlled multi-centre trial

**Authors:**

Ravi K Menon (menon.ravik@gmail.com)  
Anna P Rickard (anna.rickard@mrc-epid.cam.ac.uk)  
Nasima Mannan (n.mannan@qmul.ac.uk)  
Peter M Timms (peter.timms@homerton.nhs.uk)  
Stephen J Sharp (stephen.sharp@mrc-epid.cam.ac.uk)  
Adrian Martineau (a.martineau@qmul.ac.uk)  
Barbara J Boucher (bboucher@doctors.org.uk)  
Tahsin A Chowdhury (Tahseen.Chowdhury@bartsandthelondon.nhs.uk)  
Christopher J Griffiths (c.j.griffiths@qmul.ac.uk)  
Simon J Griffin (simon.griffin@mrc-epid.cam.ac.uk)  
Graham A Hitman (g.a.hitman@qmul.ac.uk)  
Nita G Forouhi (nita.forouhi@mrc-epid.cam.ac.uk)

**Version:** 2 **Date:** 29 November 2012

**Author's response to reviews:**

Changed author details as they had some spelling mistakes and differences in the order of authorship from the manuscript. Also uploaded the ethics permissions for the trial.